Author's response to reviews

Title: Determinants of patient satisfaction in ambulatory oncology: a cross sectional study based on the OUT-PATSAT35 questionnaire

Authors:

Thanh Vân France Nguyen (tvfnguyen@chu-besancon.fr)
Jean-François Bosset (jean-francois.bosset@univ-fcomte.fr)
Alain Monnier (moma25@orange.fr)
Valérie Perrin (vperrin@chbm.fr)
Cédric Bauman (cedric.baumann@medecine.uhp-nancy.fr)
Anne Brédart (anne.bredart@curie.net)
Mariette Mercier (mariette.mercier@univ-fcomte.fr)
Jacqueline Fournier (tvfnguyen@chu-besancon.fr)

Version: 5 Date: 16 December 2011

Author's response to reviews: see over
16th December, 2011

Dear Editor

You will find attached the revised paper entitled “Determinants of patient satisfaction in ambulatory oncology: a prospective study based on the OUT-PATSAT35 questionnaire” for submission: all changes have been highlighted in the text.

Here is a point-by-point response to the queries:

1. Please provide an e-mail for Jacqueline Fournier.
   jacfou@orange.fr

2. Please note that the 'authors contributions' and 'Acknowledgements' were in the wrong place. I have inserted the necessary headings and replaced. Please check. It’s correct.

3. Please check the presentation of table 1.
   Changes have been made to table 1.

4. There is a reference cited for [123], however there is no such reference. please clarify.
   The right reference is [23].

5. It is our house style to have all Additional files cited within the main body of text, please make the necessary citations.
   The 2 additional files have been cited in the paragraph study procedures and measures.

Thank you for your consideration,
Yours sincerely,
Merry Christmas and happy new year

On behalf of the authors
F. NGUYEN M.D.